Author’s response to reviews

Title: PREVALENCE AND RISK FACTORS FOR FRAILTY AMONG HOME CARE CLIENTS

Authors:

Irma Nykänen (Irma.Nykanen@uef.fi)
Minna Miettinen (minna.miettinen@kotikone.fi)
Miia Tiihonen (miia.tiihonen@uef.fi)
Sirpa Hartikainen (sirpa.hartikainen@uef.fi)

Version: 2 Date: 06 Sep 2017

Author’s response to reviews:

Dear Editor,

I am happy to know that our manuscript “Prevalence and risk factors for frailty among home care clients” is potentially acceptable for publication in BMC Geriatrics.

The citation 37 has dropped and citation 38 has been changed to 37.

On behalf of the author team

Yours sincerely,

Irma Nykänen, Adjunct Professor
Institute of Public Health and Clinical Nutrition
School of Medicine
University of Eastern Finland
Email: Irma.Nykanen@uef.fi